Author’s response to reviews

Title: Association of nutritional support programs with zinc deficiency in Colombian children: a cross-sectional study

Authors:
Angela Maria Maria Pinzon-Rondon (angela_pinzon@hotmail.com)
Alfonso Hoyos-Martínez (alfonsohoyos@gmail.com)
Daniela Parra-Correa (danielaparrac_93@hotmail.com)
Ana Maria Pedraza-Flechas (anama.pedraza@urosario.edu.co)
Angela Maria Ruiz-Sternberg (angela.ruiz@urosario.edu.co)

Version: 2 Date: 25 Jul 2019

Author’s response to reviews:

Dear BMC Nutrition editors,

Thank you and the reviewers for your help preparing our manuscript. We have uploaded a final version answering to your recommendations as follows:

1. Funding
   -- In the Funding section, we have made clear that the funding body did not have any role in the development of the manuscript. Page 12, lines 49 to 54.

2. Authors' contributions
   -- We replaced the names of the authors for their full initials. Page 13, lines 4 to 12.

3. Clean manuscript
   -- We have checked the manuscript and tables and we have uploaded a clean version of the manuscript. Please let us know if you required anything else.
Sincerely,

Angela M Pinzon-Rondon
Ana M Pedraza-Flechas